Author's response to reviews

**Title:** Using high titer West Nile intravenous immunoglobulin from selected Israeli donors for treatment of West Nile virus infection

**Authors:**

David Ben-Nathan (davidbn@013.net.il)  
Orly Gershoni-Yahalom (orlyge@bgu.ac.il) 
Itzchak Samina (isami@moag.gov.il)  
Yevgeny Khinich (vaccine@moag.gov.il)  
Israel Nur (israel@omrix.co.il)  
Orgad Laub (orgad@omrix.co.il)  
Ahuva Gottreich (ahuva@sheba.health.gov.il)  
Michael Simanov (vaccine@moag.gov.il)  
Angel Porgador (angel@bgu.ac.il)  
Bracha Rager-Zisman (rager@netvision.net.il)  
Nadav Or (nadav@omrix.co.il)

**Version:** 6  **Date:** 27 January 2009

Author's response to reviews:

27.1.2009

Bio Med Central Editor

BMC Infectious Diseases

Re: MS 8846492292163447

Dear Editor,

Manuscript (MS: 8846492292163447) entitled "Using high titer West Nile Intravenous immunoglobulin from selected Israeli donors for treatment of West Nile virus infection."

Following the editor's comment

We revised as follows:

Competing interests
DBN, OGY, IS, YK, AG, MS, AP and BRZ declare that they have no competing interests. NO, IN and OL are employees of OMRIX Biopharmaceuticals.

Sincerely

Dr. David Ben Nathan

Corresponding author